Author's response to reviews

Title: The STRIPES Trial - Support to Rural India's Public Education System

Authors:

Alex Eble (ae@effint.org)
Vera Mann (vera.mann@lshtm.ac.uk)
Preetha Bhakta (preetha@naandi.org)
Rashmi Lakshminarayana (rln@effint.org)
Chris Frost (chris.frost@lshtm.ac.uk)
Diana Elbourne (diana.elbourne@lshtm.ac.uk)
Peter Boone (pb@effint.org)

Version: 2 Date: 6 August 2009

Author's response to reviews:

We were informed by two organisations involved in randomised trials in education in India (Educational Initiatives Pvt. Ltd., Ahmedabad and Educational Resource Unit, ERU Consultants Pvt. Ltd., Delhi) that no further approval is needed. inserted into Ethical approval session of the manuscript.